Author's response to reviews

Title: Primary Central Nervous System Angiosarcoma: A Case Series

Authors:

James R Hackney (jhackney@uab.edu)
Cheryl A Palmer (capalmer@uab.edu)
Kristen O Riley (koriley@uabmc.edu)
Joel K Cure (jcure@uabmc.edu)
Hassan M Fathallah-Shaykh (hfathall@uab.edu)
Louis B Nabors (bnabors@uab.edu)

Version: 4 Date: 29 February 2012

Author's response to reviews: see over
Dear Casy:

On behalf of my coauthors, I am re-submitting our case series, “Primary Central Nervous System Angiosarcoma : A Case Series,” for consideration for publication in *Journal of Medical Case Reports*.

At your request, I have divided the Case Presentation subsection in the abstract into two cases [Case #1 and Case #2] and added a few more details about the cases. I hope this is acceptable to you.

If this is not what you wished for me to do, please let me know.

Thank you very much for considering our revised manuscript. If you have any additional questions or concerns, please do not hesitate to contact me.

Sincerely yours,

Cheryl Ann Palmer, M.D.
Division of Neuropathology
University of Alabama at Birmingham
1960 6th Avenue South; PD6A 175E
Birmingham, AL 35294
(205) 934-2164 – phone
(205) 975-7548 - fax
Email: capalmer@uab.edu